CIS 500 Software Foundations (Fall 2002) Homework Assignment 4

Lambda Calculus Due: Thursday, October 3, 2002, by 4PM Submission instructions:

• Turn in your solutions in hardcopy form to Christine Metz in 556 Moore.

• If it is difficult for you to come to campus on Thursday, you may submit your solutions by fax to 215 898-0587. You must also email Christine Metz (cmetz@cis.upenn.edu) to alert her to look for your fax.

1 Exercise 5.2.4 in TAPL. 2 Exercise 5.2.5 in TAPL. 3 Exercise 5.2.7 in TAPL. 4 Exercise 5.2.8 in TAPL. 5 Exercise 5.2.11 in TAPL.

All of the solutions can be found in the back of the book.